Author's response to reviews

Title: Preoperative Planning of Calcium Deposit Removal in Calcifying Tendinitis of the Rotator Cuff - Possible Contribution of Computed Tomography, Ultrasound and Conventional X-Ray

Authors:

Kaywan Izadpanah (izadpanah@me.com)
Martin Jaeger (martin.jaeger@uniklink-freiburg.de)
Dirk Maier (dirk.maier@uniklink-freiburg.de)
Norbert P Südkamp (norbert.sudkamp@uniklink-freiburg.de)
Peter Ogon (peter@ogon.ws)

Version: 7 Date: 19 October 2014

Author's response to reviews:

Dear Editor,

The following revisions have been performed.

- Please include the full name of the ethics board that approved the study in the methods section.

Response:

The present study was approved by the local ethics committee of the University Medical Center Freiburg (Nr.: 329/12).